Author's response to reviews

Title: Planning, timing and quit success: Study protocol of a randomised controlled trial.

Authors:

Ron Borland (Ron.Borland@cancervic.org.au)
James Balmford (James.Balmford@cancervic.org.au)
Elena Swift (Elena.Swift@cancervic.org.au)

Version: 4 Date: 28 December 2012

Author's response to reviews:

Dear Editors,

As discussed by email, we have added a sentence to the manuscript directing interested readers to a newly-established website on which they can view the interventions offered in each study condition.

Kind regards,

James Balmford Ph.D (on behalf of all authors)